Author's response to reviews

Title: A large-scale assessment of hand hygiene quality and the effectiveness of the "WHO 6-steps"

Authors:

Laszlo Szilagyi (lalo@ms.sapientia.ro)
Tamas Haidegger (haidegger@eestec.hu)
Akos Lehotsky (akos.lehotsky@gmail.com)
Melinda Nagy (meliole28@gmail.com)
Erik A Csonka (nyari_mikulas@yahoo.com)
Xiuying Sun (xiuying_sun@nuhs.edu.sg)
Kooi L Ooi (kooi_li_ooi@nuhs.edu.sg)
Dale A Fisher (dale_andrew_fisher@nuhs.edu.sg)

Version: 2 Date: 6 June 2012

Author's response to reviews:

Confidential comments to the Editors were added to the submission.